Author's response to reviews

Title: The effectiveness of exercise as a treatment for postnatal depression: Study protocol

Authors:

Amanda J Daley (a.daley@bham.ac.uk)
Kate Jolly (c.b.jolly@bham.ac.uk)
Debbie J Sharp (Debbie.Sharp@bristol.ac.uk)
Katrina M Turner (katrina.turner@bristol.ac.uk)
Ruth V Blamey (r.v.blamey@bham.ac.uk)
Sarah Coleman (s.j.coleman.1@bham.ac.uk)
Mary McGuinness (Mary.McGuinness@bsmhft.nhs.uk)
Andrea K Roalfe (a.k.roalfe@bham.ac.uk)
Ian Jones (jonesir1@cf.ac.uk)
Christine MacArthur (c.macarthur@bham.ac.uk)

Version: 2 Date: 6 February 2012

Author's response to reviews:

Corrected reference 19.
Provided evidence of funding.